Author’s response to reviews

Title: Impact of smoking and smoking cessation on overweight and obesity: Scotland-wide, cross-sectional study on 40,036 participants.

Authors:

Daniel F Mackay (daniel.mackay@glasgow.ac.uk)
Linsay Gray (L.gray@sphsu.mrc.ac.uk)
Jill P Pell (jill.pell@glasgow.ac.uk)

Version: 6  Date: 22 March 2013

Author’s response to reviews:

We have addressed all five minor revisions requested by the referees. There were no major revisions.

1. be has been replaced by been - line 89
2. overweight was changed to overweight/obesity - line 255
3. the abstract now states that height and weight were measured by a trained interviewer and not self-reported
4. the abstract now states representative of the scottish population >=16 years
5. figure 2 footnote marked *amended as requested